Author's response to reviews

Title: Evidence based guidelines for use of probiotics in preterm neonates

Authors:

Girish C Despande (girish.deshpande@swahs.health.nsw.gov.au)
Shripada C Rao (shripada.rao@health.wa.gov.au)
Anthony D Keil (tony.keil@health.wa.gov.au)
Sanjay K Patole (sanjay.patole@health.wa.gov.au)

Version: 9 Date: 14 July 2011

Author's response to reviews:

Date: 14/07/2011

The Editor,
BMC Pediatrics
Dear Sir/Madam
Ref: MS: 3095885148532319: Evidence based guidelines for use of probiotics in preterm neonates. Girish Deshpande, Shripada Rao, Anthony Keil and Sanjay Patole

We are now submitting the reformated version of the manuscript as requested.

Kind Regards,
Prof Sanjay Patole, MD, DCH, FRACP, DCT, DrPH
Department of Neonatal Paediatrics, KEM Hospital for Women
Bagot Road, Subiaco, City of Perth
Western Australia 6008
Tel: 08-93401260, Fax: 08-93401266
E-mail sanjay.patole@health.wa.gov.au